Author’s response to reviews

Title: Waist height ratio predicts chronic kidney disease: A systematic review and meta-analysis

Authors:

Ling Liu (pql599101725@163.com)
Yanqiu Wang (15555337667@163.com)
Wanjun Zhang (wjzhang0207@163.com)
Weiwei Chang (xiaowei8601@163.com)
Yuelong Jin (94017416@qq.com)
Yingshui Yao (yingshuiyao@126.com)

Version: 2 Date: 12 Nov 2019

Author’s response to reviews:

Part A (Editorial comments)

Comments:

① Add the period covered by the review in the title.

② The title of each table, graph should refer to the study. Please add, if not already in the table/graph this sentence to current the titles: A systematic review and meta-analysis on the waist height and chronic kidney disease (1998-2019).

Reply:

Thank you for your suggestion, all your suggestions are very important, they have important guiding significance for my thesis writing and research work!

① We have revised the title and added the study period after the title.

② We have add each table/graph with the titles: A systematic review and meta-analysis on the waist height and chronic kidney disease (1998-2019).
Many grammatical and typographical errors have been revised. All the lines and pages indicated above are in the revised. Thank you and all reviewers for the kind advice. We hope that the revision is acceptable and look forward to hearing from you soon. If you have any question about this paper, please don’t hesitate to let me know.